Author's response to reviews

Title: Evaluation of Wound Healing and Invivo anti-inflammatory Activity of Rhizomes of Rumex abyssinicus J. in Mice

Authors:

Eshetu Mulisa (mulisa2e@yahoo.com)
Kaleab Asres (kasres@gmail.com)
Ephrem Engidawork (ephrem.engidawork@aau.edu.et)

Version: 3
Date: 3 December 2014

Author's response to reviews:

To
Editor-in-chief
BMC Complementary and Alternative Medicine
Dear Sir,


We have already sent a manuscript entitled “Evaluation of Wound Healing and Invivo anti-inflammatory Activity of Rhizomes of Rumex abyssinicus J. in Mice” for publication in your BMC Complementary and Alternative Medicine. And revised the paper based on the questions.

1. Submission and approval was stated
   • Line number 126-127

2. Line numbers are added